Author's response to reviews

Title: Hospital-based, prospective, multicentre surveillance to determine the incidence of intussusception in children aged below 15 years in Germany

Authors:

Nicolai Bissantz (nicolai.bissantz@rub.de)
Andreas C Jenke (andreas.jenke@helios-kliniken.de)
Matthias Trampisch (matthias-trampisch@amib.rub.de)
Renate Klaassen-Mielke (renate.klaassen-mielke@rub.de)
Kathrin Bissantz (kathrin.bissantz@web.de)
Hans-Joachim Trampisch (trampisch@amib.ruhr-uni-bochum.de)
Tim Holland-Letz (holland-letz@amib.rub.de)

Version: 5 Date: 22 February 2011

Author's response to reviews:

Major revisions

Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend.

No minor revisions were taken care of. We kindly request that you carry on the remaining revise requests.